Author's response to reviews

Title: Assessment of clinical competencies using clinical images and videos

Authors:

Nabil D Sulaiman (nsulaiman@sharjah.ac.ae)
Hossam Hamdy (hamdy@sharjah.ac.ae)

Version: 3 Date: 17 October 2012

Author's response to reviews: see over
13 OCT 2012

Dear editor

BMC Medical Education Journal

Dear Sir/ Madam

Re: Submitting manuscript for publication in BMC Medical Education

The authors would like to submit the article “Assessment of clinical competencies using clinical images and videos” for publication in BMC Medical Education. The manuscript is relevant to the scope of the journal. It addresses an important area of medical education - assessment of clinical competency using a cost effective method.

The manuscript is not under review by any journal and the two authors declare No conflict of interest.

The study was approved by the University of Sharjah Research and Ethics Committee.

Sincerely yours

Nabil Sulaiman

Principle Investigator